Author's response to reviews

Title: Childhood autism in a 13 year old boy with oculocutaneous albinism: a case report

Authors:

Muideen O Bakare (mobakare2000@yahoo.com)
Nkeiruka N Ikegwuonu (nkeiruikegwuonu@yahoo.com)

Version: 12 Date: 20 February 2008

Author's response to reviews: see over
Dear Editor,

**Re: Childhood autism in a 13 year old boy with oculocutaneous albinism: a case report**

The manuscript has been modified as requested.

Lower case has been used in “Case presentation” as requested.

The Figures have also been cited in the manuscript text.

Thank you.

Yours faithfully,

**Dr. Muideen Owolabi Bakare, M.B.B.S, FMCPsych**
(Corresponding Author)